Author's response to reviews

Title: Acupuncture as a treatment for functional dyspepsia: Design and methods of a randomized controlled trial

Authors:

Hui Zheng (zhui126@yahoo.com.cn)
Xiao-ping Tian (ttxp2006@126.com)
Ying Li (jialee@mail.sc.cninfo.net)
Fan-rong Liang (acuresearch@126.com)
Shu-guang Yu (ysg2858@yahoo.com.cn)
Xu-guang Liu (lxg428@sohu.com)
Yong Tang (tangyongcn2001@yahoo.com.cn)
Xu-guang Yang (yxg_61@yahoo.com.cn)
Jie Yan (yi5381159@yahoo.com.cn)
Guo-jie Sun (sunguojie@sina.com)
Xiao-rong Chang (xrchang1956@163.com)
Hong-xing Zhang (zhxzj99@yahoo.com.cn)
Ting-ting Ma (matingtingcn@yahoo.com.cn)
Shu-yuan Yu (547025100@qq.com)

Version: 5 Date: 31 July 2009

Author's response to reviews: see over
Dear editors,
Thanks for the comments and advice. As you suggested, we asked the English Manager Science Editing (http://www.sciencemanager.com/) for copyediting service. All changes were highlighted using the track changes option in Microsoft Word.

If there are still any questions, please do not hesitate to contact us.

Sincerely yours,
Fan-rong Liang
Tel.: 028- 66704320
Fax: 86-28-87784606
E-mail: acuresearch@126.com